Author's response to reviews

Title: Health differentials in the older population of England: An empirical comparison of the materialist, lifestyle and psychosocial hypotheses.

Authors:

George B Ploubidis GP (George.Ploubidis@LSHTM.ac.uk)
Bianca L DeStavola BS (bianca.destavola@lshtm.ac.uk)
Emily Grundy EG (Emily.Grundy@LSHTM.ac.uk)

Version: 5 Date: 29 March 2011

Author's response to reviews:

Dear Editors,

we have edited the manuscript as requested. We are delighted that it is now accepted for publication at BMC Public Health